Author's response to reviews

Title: Social inequalities in self-rated health by age: Cross-sectional study of 22,457 middle-aged men and women

Authors:

Emily McFadden (ecm33@cam.ac.uk)
Robert Luben (robert.luben@phpc.cam.ac.uk)
Shelia Bingham (sab@mrc-dunn.cam.ac.uk)
Nicholas Wareham (njw1004@medschl.cam.ac.uk)
Ann-Louise Kinmonth (alk25@medschl.cam.ac.uk)
Kay-Tee Khaw (kk101@medschl.cam.ac.uk)

Version: 4 Date: 4 July 2008

Author's response to reviews:

COMMENT:
References - The reference list should contain all authors¿ names, if there are 30 or below. The term ¿et al.¿ should not be used.

RESPONSE:
Thank you. We have re-checked the reference list and all authors are now listed for all references.